Author's response to reviews

Title: ORCHID: Open reduction and internal fixation versus casting for highly comminuted intra-articular fractures of the distal radius - A randomized clinical multi-center trial A randomized clinical multi-center trial

Authors:

Christoph Bartl (chbartl@gmx.de)
Dirk Stengel (zkf@ukb.de)
Thomas Bruckner (bruckner@imbi.uni-heidelberg.de)
Inga Rossion (inga.rossion@med.uni-heidelberg.de)
Steffen Luntz (Steffen.Luntz@med.uni-heidelberg.de)
Christoph Seiler (christoph.seiler@med.uni-heidelberg.de)
Florian Gebhard (Florian.Gebhard@uniklinik-ulm.de)

Version: 7 Date: 2 March 2011

Author's response to reviews:

cover letter 2.3.2011:

Dear Prof. Goldsmith,

the abstract is revised and structures in the sections, background, methods/design and discussion.

We hope to meet now the criterias for acceptance of the manuscript.

thank you very much for your help

sincerely

christoph bartl